# RESEARCH Open Access

# Effects of family-based treatment on adolescent outpatients treated for anorexia nervosa in the Eating Disorder Unit of Helsinki University Hospital

Svetlana Oshukova<sup>1\*</sup>, Jaana Suokas<sup>1</sup>, Mai Nordberg<sup>2</sup> and Monica Ålgars<sup>1,3,4</sup>

#### **Abstract**

**Background** Family therapy for adolescent anorexia nervosa (AN) has stronger evidence of efficacy in comparison with individual therapy, and family-based treatment (FBT) is the most evaluated in numerous randomized clinical trials. However, few studies have focused on how FBT performs outside of research settings. The current study is the first to assess clinical outcomes of FBT for adolescent AN in Finland, in a specialized outpatient clinic.

**Aim** The naturalistic outcome of outpatient FBT for adolescent AN was investigated.

**Methods** Fifty-two female patients and their families who received FBT at a tertiary eating disorders unit participated in the study. Data on their pre-treatment parameters, treatment details, and condition at the end of treatment (EOT) was collected from their medical records.

**Results** At EOT, a majority (61.5%) had achieved a full weight restoration [percentage of expected body weight (%EBW)  $\geq$  95%]. Participants with an %EBW  $\geq$  95 at EOT had a significantly higher pre-treatment %EBW than those with an EBW < 95% at EOT. Participants with an EBW  $\geq$  95% at EOT showed significantly higher total weight gain during the treatment period, a higher rate of regular menstrual periods at EOT, significantly lower rates of dietary restrictions, and less cognitive or behavioral symptoms of the eating disorder overall, compared to participants who did not achieve a normal body weight. In 22 cases (42.3%), there was no need for further treatment at the end of FBT. Participants who needed further treatment after FBT, compared to those who did not, showed significantly higher rates of psychiatric comorbidity, history of mental health treatment, and need for psychopharmacological treatment.

**Conclusions** In this naturalistic study, and in line with previous studies, FBT for AN appeared to be an effective and sometimes sufficient intervention, especially for patients with milder weight deficit and less severe psychiatric comorbidities. The results show that FBT can be successfully implemented in Finland and suggest that training more ED clinicians in FBT would be beneficial.

*Trial registration*: The study was retrospectively registered on February 8th, 2023, in ClinicalTrials.gov Protocol Registration and Results System, identifier: NCT05734573.

Keywords Adolescents, Anorexia nervosa, Family-based treatment, Outpatient

\*Correspondence: Svetlana Oshukova svetlana.oshukova@hus.fi Full list of author information is available at the end of the article



# **Plain English summary**

Family-based treatment (FBT), a form of family therapy for eating disorders (ED) in young patients, has proven to be effective in numerous large-scale studies. This study aims to evaluate outcomes of the treatment for anorexia nervosa (AN) in a 'real-world' setting in a specialized outpatient clinic in Helsinki, Finland. Fifty-two young female patients seen for treatment between June 2013 and December 2017 were included in the study. Demographic and treatment characteristics, weight, and ED symptoms before and after treatment are reported. At the end of treatment, most patients had a good outcome, including normalization of their body weight and reduced ED symptoms. The majority of patients who needed further treatment after FBT suffered from an additional psychiatric disorder together with AN.

# **Background**

Eating disorders (EDs) are serious psychiatric disorders that can affect individuals of all ages and genders, but female adolescents and young adults are particularly at risk. In addition to eating behavior disturbances, the symptomatology includes mental, physical, and social impairment. EDs are associated with severe medical and psychological consequences, such as osteoporosis, growth retardation, and developmental delay in those underage [1]. Anorexia nervosa (AN) is a psychiatric disorder characterized by significantly low weight, dietary restrictions, compensatory behaviors, such as compulsive exercising and purging or laxative abuse, as well as an intense fear of weight gain and a distorted body image. Early detection and referral to treatment can reduce the length of illness and improve the prognosis remarkably [2, 3].

For adolescent AN, family therapy has stronger evidence of efficacy than individual therapy, according to systematic reviews [4, 5]; it is recommended for adolescent AN in several clinical guidelines [6, 7].

Among the family-focused treatment modes, familybased treatment (FBT) has gained the most evidence of efficacy [8]. FBT, developed in London, emphasizes parental support to the child in the process of weight restoration and normalization of eating patterns [9]. The treatment consists of three phases. Phase I, which includes an in-session family meal, focuses on weight restoration and assisting and empowering the parents in taking responsibility for their child eating. In Phase I, the therapeutic discussions focus on the management of food and eating behaviors. Phase I continues until there is steady weight restoration and the ED symptoms begin to recede. In Phase II, age-appropriate control of eating is gradually transitioned back to the adolescent. The parents are supported in managing ED symptoms until the patient can eat well on their own. The relationship between adolescent developmental issues and the ED is also explored. In Phase III, the focus lies on establishing a healthy, adolescent identity and ending treatment well. Adolescent issues are reviewed and explored, problem solving modeled, and future problems planned for [9].

FBT has been rigorously studied in randomized clinical trials (RCT) in Australia, Canada, and the United States [10-18]. The proportion of patients achieving a good clinical improvement (i.e.,  $\geq$  85% of ideal body weight) in FBT trials varies across studies, ranging between 34 and 90% [19]. FBT response, defined as an improvement in weight and eating-related psychopathology, is reported to be at an average of about 75% (range 60–85%) [20].

While research trials have shown evidence of the efficacy of FBT, there is still a lack of studies providing evidence of the treatment's effectiveness when delivered in routine practice settings, as well as within different national and cultural settings. Naturalistic studies reporting treatment effects in routine practice settings in different cultures and types of care would confirm the generalizability of the treatment. It would thus encourage clinicians across settings to implement the treatment. Though studies on how FBT performs in non-research settings are still few [21-25], the results of those are promising; FBT seems to be not only efficacious but also an effective treatment for adolescents with AN. A growing body of real-life study reports contributes to the evidence base for FBT as a first-line treatment method. Up to now, clinical outcomes of FBT in Finland have not been reported, and the current study aims at filling this gap.

In assessing the clinical outcomes of treatments for AN, remission is usually defined as achieving a normal body weight ( $\geq$ 95% of ideal body weight) and a reduction in eating-related symptoms, both cognitive and behavioral, to subclinical levels [15, 26]. Still, it is not easy to predict the outcome of FBT in clinical practice, as only few parameters have shown potential influence on the response to the treatment. However, early weight gain, mostly defined as a gain of 2–3 kg during the first month of treatment, has been shown to be a prognostic indicator of remission in FBT [17, 26–29].

Comorbid psychiatric disorders, the age of 18 and over, prior hospitalization, longer illness duration, as well as high parental expressed emotion (a measure of caregiver attitudes toward a relative with an illness), have been associated with lower remission rates [30, 31]. Bulimic

symptoms and emotional dysregulation have been found to predict higher levels of ED symptoms post-treatment [32]. Comorbid psychiatric disorders have been shown to predict a greater chance of dropping out from FBT [31].

There are still few published reports on FBT in Northern European countries [32–34].

The primary aim of this study was to explore the effects of FBT in a naturalistic setting, in a sample of adolescents with AN referred to the Eating Disorder Unit of Helsinki University Hospital. We hypothesized that FBT would be effective in restoring body mass and reducing ED symptoms, and a sufficient intervention for some of the treated patients. We were interested in whether the results of FBT could be predicted by some pre-treatment characteristics of the patients or by the weight gain rate during the treatment. In line with previous research findings [30, 31], we hypothesized that comorbid psychiatric disorders, older age, and longer illness duration would be associated with poorer treatment outcomes. Further, we wanted to look for common features in adolescents who were still in need of psychiatric treatment after FBT.

#### Methods

# Study design

The study was conducted at the outpatient department of the Eating Disorder Unit, which is a tertiary health care unit at Helsinki University Hospital's Psychiatry Department. The unit provides specialized assessment and treatment for adolescents and adults with severe EDs in the Helsinki region, with around 1.6 million inhabitants. At the unit, treatment is primarily provided at an outpatient level, but day patient and inpatient treatments are also available. Patients and their guardians received information about the study both orally and in a cover letter. The participants were assured of the confidentiality and anonymity of the data and of the voluntary nature of participation. The patients and their guardians gave written confirmation of their consent. Data for the study was gathered from the medical records of the patients at the end of treatment.

# **Participants**

The participants and their parents were recruited from June 1, 2013, through December 31, 2017. All outpatients aged 13–18 years meeting the diagnostic criteria for AN and admitted to FBT at the Eating Disorder Unit of Helsinki University Hospital were invited to participate in the study. During the study period, 428 adolescent patients started outpatient treatment at the Eating Disorder unit.

Of those, 52 female patients diagnosed with AN, with a mean age of 14.50 (SD=1.31) years, were admitted to FBT, and all of them participated in the study.

The pre-treatment parameters for the sample are presented in Table 1. The mean percentage of expected body weight (%EBW) at the beginning of treatment was 83.08 (SD=8.58). Five patients had an EBW  $\geq$  95%; at pretreatment. As these adolescents still had a weight deficit in comparison to their weight before the ED onset, and presented clinically similarly to those more underweight, they were also included into the study sample. On average, the participants had suffered from ED symptoms for 13.29 (SD=8.15) months before starting FBT.

#### Intervention

At the Eating Disorder Unit, the aim is to offer FBT as the first-line treatment for adolescent patients with AN. However, due to the often severe symptomatology of the patients treated at the unit, as well as limited resources, only part of the patients can be admitted to FBT. On the patient's first visit to the unit, the attending psychiatrist usually makes a clinical decision on whether FBT is suitable for the patient. The decision is based on an assessment of the patient's medical and mental state, likelihood to commit to the treatment, and family functioning. At the time of the data collection, FBT at the unit was carried out by a family therapist experienced in the treatment of EDs, who had received certified FBT training at the Maudsley Hospital in London. Treatment fidelity was ensured by regular supervision of case material by a qualified supervisor. The duration of FBT was commonly 6–12 months, in most cases comprising 10–20 sessions. At the beginning of treatment, there were weekly sessions; later in treatment, the sessions were less frequent.

**Table 1** Pretreatment characteristics of 52 adolescent girls administered to FBT for AN

Pre-treatment parameter	N=52
Age (years), mean(SD)	14.50 (1.31)
Living in an intact family, n(%)	41 (78.8%)
Living with a single parent, n(%)	5 (9.6%)
%EBW, mean(SD)	83.08 (8.58)
Duration of the eating disorder (months), mean(SD)	13.29 (8.15)
Obsessive exercise, n(%)	32 (61.5%)
Psychiatric comorbidity, n(%)	14 (26.9%)
Depression	6 (11.5%)
Anxiety disorder	3 (5.8%)
Obsessive-compulsive disorder	4 (7.7%)
Conduct disorder	1 (1.9%)
History of previous treatment for AN, n(%)	17 (32.7%)
Psychopharmacological treatment, n(%)	15 (28.8%)

The patient and their family met with their psychiatrist every four to six weeks, and the family therapist also took part in these meetings. The treatment was implemented in accordance with the FBT treatment manual [9], but in a few cases, the FBT protocol was modified to have an increased number of sessions and longer duration due to the patient's clinical condition.

### **Treatment characteristics**

The number of FBT treatment sessions ranged from 5 to 40 ( $M\!=\!15.3$ ,  $SD\!=\!7.34$ ). In 11 cases (21.2%), the treatment demanded more than 20 sessions. The treatment lasted up to 23 months, with an average duration of 10.25 ( $SD\!=\!5.0$ ) months. In all cases but one, FBT was finished in accordance with the clinical situation and treatment plan, and only one family dropped out of treatment. This withdrawal happened after session 10 of FBT when the family decided to continue the treatment at a private clinic. As the participant had participated in half of the planned FBT sessions before dropping out and had achieved change in weight and ED symptoms, we did not exclude her from the study.

## **Definition of disorders**

In Finland, EDs are diagnosed in accordance with the International Classification of Diseases (ICD-10) [35]. The diagnostic criteria differ between the *Diagnostic and Statistical Manual of Mental Disorders, Fifth Edition*, (DSM-5) and ICD-10, so that the definition for EDs in DSM-5 is broader [6]. Thus, in line with previous Finnish studies [36, 37], we used a broader definition for AN. Broad AN includes ICD-10 AN (F50.0) and ICD-10 atypical AN (F50.1). An atypical AN diagnosis is used for patients in whom one or more key features of AN, such as amenorrhea or weight loss below 85% of expected, is absent, but who otherwise present a typical clinical picture.

# **Assessments**

Pre-treatment information about the patients' growth and health before ED onset, as well as diagnostic evaluations of comorbid psychiatric disorders, was gathered from their medical records. Assessment at baseline and at the end of treatment (EOT) included weight, height, ED symptoms, and psychopathology. A psychiatrist performed a rigorous clinical review, which included a non-structured interview alone with the patient, as well as with the whole family.

Validated measures were offered before FBT to all the patients for assessing ED symptoms (Eating Disorder Examination Questionnaire; EDE-Q [38]), depressive symptoms (Beck Depression Inventory; BDI [39]), and psychosocial functioning (The Clinical Impairment Assessment Questionnaire; CIA [40]).

EDE-Q, BDI, and CIA scores were only available for a small number of participants, either due to the participant not having returned them or the score not having been recorded in their medical records. Therefore, these scores were not included in this study.

According to FBT protocol, weight was assessed before every treatment session. However, only the results of the weighing at session 5 were included in the study as a marker of early response, in line with previous research [10, 28]. The participants were weighed in their underwear on a balance beam scale that was regularly re-calibrated. Height was also regularly measured.

The percentage of the expected body weight (%EBW) was calculated using national charts for age, height, and gender, according to the 50th percentile body mass index (BMI). Weight-related remission was defined as EBW  $\geq$  95%; partial remission was considered when EBW  $\geq$  85%, but < 95%; patients with EBW < 85% were considered not recovered.

The attending psychiatrists' reports of ED symptoms and psychopathology at baseline and at EOT were used. The participants were asked about menstrual period regularity, daily meal schedule, dietary restrictions or special diets, obsessive exercise or other compensative behaviors, body-related feelings, mood symptoms, and anxiety, and the psychiatrist recorded their answers. The participants' caregivers, as well as their therapist, were asked to express their opinion regarding whether the patient benefited from FBT or not, at the last FBT session, and this information was available from the medical records.

# **Statistics**

The statistical analyses were performed using IBM SPSS Statistics 25. First, the data was analyzed for descriptive parameters and incidence rates of all the variables. According to skewness and kurtosis, the data was mostly normally distributed. To assess treatment outcomes, the change in %EBW was tested using a two-way analysis of variance for repeated measures, and the McNemar chisquare test was used to analyze the change in the incidence rates of symptoms. Further, for comparing groups of patients at the EOT, Student's t-test and chi-square tests were used. All analyses were two-tailed. The alpha level was set at p < 0.05. We also calculated Cohen's d to estimate the effect sizes of statistically significant differences in the *t*-test, interpreting an effect size of 0.2 to 0.5 as small, 0.5 to 0.8 as medium, and over 0.8 as large. For the chi-square test, the phi  $(\phi)$  coefficient was a measure for the effect size. A value of 0.1 was considered a small effect, 0.3 a medium effect, and 0.5 a large effect [41].

#### **Results**

The findings from treatment outcomes are exhibited in Table 2. At EOT, 32 patients (61.5%) had achieved full weight restoration (>95% EBW), 16 (30.8%) achieved partial weight-based remission (85 < %EBW < 95), and only in four (7.7%) cases EBW remained below 85%. The mean %EBW at EOT was 98.65 (10.96).

From the parents' point of view, as well as assessed by the therapist, 45 patients (88.2%) benefited from the treatment. At EOT, there was a statistically significant increase in the incidence of regular menstrual periods and regular meals, and a significant decrease in obsessive exercise and the proportion of participants with dietary restrictions. At EOT, 78.8% of patients reported no fear of gaining weight, 75% reported no appearance dissatisfaction, and 98.5% reported no body shape distortion.

In 30 cases (57.7%), participants continued treatment after FBT either at the Eating Disorder Unit of Helsinki University Hospital (25%, n=13) or at an adolescent psychiatric outpatient clinic (30.8%, n=16). A common reason for a referral to the adolescent psychiatric unit was either a comorbid psychiatric disorder or need for further, less specific psychotherapeutic support for the convalescence and age-appropriate psychological challenges. Those who continued treatment for AN at the Eating Disorder Unit, were offered a focused individual or group outpatient intervention according to their medical and psychiatric symptomatology. As

mentioned above, one family chose to continue treatment at a private clinic after ten sessions of FBT.

Table 3 shows a comparison between participants who achieved a normal body weight at EOT and participants who did not. With a large effect size, the participants with a %EBW ≥ 95 at EOT had a significantly higher pre-treatment %EBW and showed a significantly higher total weight gain during the treatment period than those who did not achieve normal body weight. Compared to those who remained underweight, those achieving normal body weight at EOT showed, with large effect sizes, significantly higher rates of a regular menstrual period and significantly lower rates of dietary restrictions and cognitive or behavioral symptoms of the ED overall at EOT.

Need for further treatment after FBT (Table 4) was slightly higher among participants with weight deficit at EOT (65% vs 53.1%). Participants who still needed treatment after FBT, compared to those who did not need further treatment, had significantly higher rates of psychiatric comorbidity, history of psychiatric treatment, and more need for psychopharmacological treatment. At EOT, those in need of further treatment had significantly higher rates of anxiety, depression symptoms, and ED symptoms than participants who did not need further treatment.

**Table 2** Outcomes of FBT in 52 adolescent girls treated for AN

Parameter	Pretreatment	At EOT	Statistical significance of the change
%EBW, mean ( <i>SD</i> )	83.08 (8.58)	98.65 (10.96)	$t(df) = 13,310(51)^*, p < 0.001, d = 1.58$
Regular menses, n (%)	1 (1.9%)	28 (54.9%)	$X^{2}(df) = 24.083(1)^{**}, p < 0.001, \varphi = 0.48$
Obsessive exercise, n (%)	32 (62.7%)	0 (0%)	$X^{2}(df) = 28.033(1)^{**}, p < 0.001, \varphi = 0.48$
Regular meals, n (%)	5 (9.8%)	47 (90.4%)	$X^{2}(df) = 39.024(1)^{**}, p < 0.001, \varphi = 0.48$
Binge eating, n (%)	3 (5.9%)	1 (1.9%)	p=0.63**
Special diets, n (%)	17 (32.7%)	7 (13.5%)	$X^{2}(df) = 8.100(1)^{**}, p < 0.001, \varphi = 0.48$

	At the EOI
85% < EBW < 95%, n (%)	16 (30.8%)
EBW ≥ 95%, n (%)	32 (61.5%)
Benefited from the FBT (parents' assessment), n (%)	45 (88.2%)
Benefited from the FBT (therapist's assessment), n(%)	45 (88.2%)
No dietary restrictions, n (%)	40 (76.9%)
No appearance dissatisfaction, n (%)	39 (75%)
No fear of gaining weight, n (%)	41 (78.8%)
No body image distortion, n (%)	46 (88.5%)
No need for further treatment, n (%)	22 (42.3%)

<sup>\*</sup> Paired samples t-test; df = degree of freedom; Cohen's d used to assess the Effect Size.\*\*McNemar Chi-Square test,  $X^2 = Chi$  Square, df = degree of freedom, sample size N = 52;  $Phi (\phi)$  was used to assess the Effect Size

**Table 3** Differences in pre-treatment characteristics, treatment details, post-treatment symptoms between those who achieved 95% of expected body weight at EOT and those who did not, in 52 female adolescents received FBT for AN

	%EBW≥95 n=32	%EBW < 95 n = 20	Statistical significance of the difference
Psychiatric comorbidity, n(%)	9 (28.1%)	5 (25.0%)	p=1.00*
Time(months) before FBT, mean (SD)	13,38 (8.86)	13.15 (7.08)	$p = 0.92^{**}$
Living in intact family, n (%)	25 (78.1%)	16 (80.0%)	$p = 1.00^*$
Previous treatment for AN,n (%)	12 (37.5%)	5 (25.0%)	$p = 0.38^*$
Psychiatric medication, n (%)	10 (31.3%)	5 (25.0%)	$p = 0.76^*$
Pre-treatment %EBW,mean (SD)	86.9 (8.61)	76.9 (3.58)	$t(df) = 4.891(50)^{**}, p < 0.001, d = 1.52$
Pre-treatment obsessive exercise, n (%)	23 (71.9%)	9 (45.0%)	$p = 0.05^*$
FBT sessions number, mean (SD)	15.75 (7.99)	14.60 (6.28)	$p = 0.56^{**}$
Early weight gain, mean (SD)	3.28 (2.71)	2.96 (1.74)	$p = 0.65^{**}$
Total weight gain,			
mean (SD)	10.71 (4.77)	6.14 (1.96)	$t(df) = 4.066(50)^{**}, p < 0.001, d = 1.25$
Regular menses at EOT, n (%)	21 (65.7%)	7 (35.0%)	$X^{2}(df) = 5.160(1)^{***}, p = 0.04, \varphi = 0.32$
Need for other treatment during FBT, n (%)	11 (34.4%)	6 (30.0%)	$p = 1.00^*$
Need for further treatment at EOT, n (%)	17 (53.1%)	13 (65.0%)	$p = 0.56^*$
Depression symptoms at the EOT, n (%)	9 (28.1%)	3 (15.0%)	$p = 0.33^*$
Anxiety at EOT, n (%)	13 (40.6%)	10 (50.00%)	$p = 0.57^*$
Dietary restrictions at the EOT, n (%)	3 (9.4%)	9 (45.0%)	$X^{2}(df) = 8.265(1)^{***}, p < 0.01, \varphi = 0.41$
No symptoms of an eating disorder at EOT, n (%)	26 (81.3%)	9 (45%)	$X^{2}(df) = 7.209(1)^{***}, p < 0.01, \phi = 0.38$

<sup>\*</sup> Fisher's Exact test; \*\* Student's Independent Samples T-Test; t = Student's t, df = degree of freedom; Cohen's d used to assess the Effect Size; \*\*\*Chi Square Test; X² = Chi Square, (df) = degree of freedom, Phi (φ) used to assess the Effect Size

**Table 4** Differences in pre-treatment characteristics, treatment details, post-treatment symptoms between those who did not need further treatment at EOT and those who needed in 52 female adolescents received FBT for AN

Parameter	No further treatment need N = 22	Further treatment need N = 30	Statistical significance of the difference
Age (years), mean (SD)	14.73(1.28	14.33(1.32)	p=0.29
Pre-treatment %EBW, mean (SD)	80.72 (7.19)	84.81 (9.20)	p = 0.09
Duration of AN (months), mean (SD)	12.59 (6.75)	13.80 (9.11)	p = 0.60
Psychiatric comorbidity, n (%)	2 (9.1%)	12 (40.0%)	$X^{2}(df) = 6.163(1)^{**}, p = 0.01 \varphi = 0.34$
Psychopharmacological treatment, n (%)	2 (9.1%)	13 (43.3%)	$X^{2}(df) = 7.250(1)^{**}, p < 0.001\varphi = 0.37$
Previous treatment for AN, n (%)	3 (13.6%)	14 (46.7%)	$X^{2}(df) = 6.293(1)^{**}, p = 0.01 \varphi = 0.35$
%EBW at EOT,mean (SD)	98.74 (9.87)	98.59 (11.86)	p = 0.96
Weight gain,mean (SD)	10.90 (4.53)	7.52 (3.99)	$t(df) = 2.844(50)^*, p < 0.001 \varphi = 0.79$
Depression symptoms at EOT, n (%)	1 (4,5%)	11 (36.7%)	$X^{2}(df) = 7.381(1)^{**}, p < 0.001\varphi = 0.38$
Anxiety at EOT, n (%)	1 (4.5%)	22 (73.3%)	$X^{2}(df) = 24.350(1)^{**}, p < 0.001 \varphi = 0.68$
Dietary restrictions at EOT, n (%)	0 (0%)	12 (40.0%)	$X^{2}(df) = 10.991(1)^{*}, p < 0.001 \varphi = 0.46$
Appearance dissatisfaction at EOT,n (%)	1 (4.5%)	12 (40.0%)	$X^{2}(df) = 9.902(1)^{*}, p < 0.001 \varphi = 0.45$
Fear of gaining weight at EOT, n (%)	0 (0%)	11 (36.7%)	$X^{2}(df) = 13.252(1)^{*}, p < 0.001 \varphi = 0.54$
Body image distortion at EOT, n (%)	0 (0%)	6 (20.0%)	$X^{2}(df) = 6.621(1)^{**}, p = 0.01\varphi = 0.38$

<sup>\*</sup> Student's Independent Samples T-Test; t = Student's t, df = degree of freedom; Cohen's d used to assess the Effect Size; \*\*Chi Square Test;  $X^2 = Chi Square$ , (df) = degree of freedom,  $Phi (\phi)$  used to assess the Effect Size

# Discussion

The aim of the present study was to evaluate the effects of FBT for adolescents with AN in terms of change in weight and ED psychopathology in a naturalistic clinical setting. Though RCT studies give clinicians guidelines

for planning care and making treatment decisions, clinical reports on treatments' naturalistic outcomes are also valuable for practitioners. To our knowledge, this is the first study on FBT for EDs in Finland, and one of very few in Northern Europe [32–34].

A majority of the participants (61.5%) achieved normal body weight (%EBW  $\geq$  95) at EOT. Across studies, reports on FBT response and remission rates range widely [19, 20, 27]. Response to FBT is evaluated to be about 75% (range 60–85%), defined as an improvement in weight and eating-related psychopathology. Thus, our results are in line with the research findings to date regarding weight restoration [19, 20]. In recent 'reallife' study reports, FBT has consistently shown its effectiveness regarding weight restoration [22–25], and our findings contribute to this evidence.

There was only one statistically significant difference in pre-treatment characteristics between those who achieved normal body weight at the end of FBT and those who did not: those who reached normal weight had a significantly higher pre-treatment %EBW than those who remained underweight. This finding is in line with previous findings showing that FBT is more effective for ED patients with less severe underweight [21, 24, 32], and that higher pre-treatment weight is a positive prognostic factor in AN patients, regardless of treatment method [21, 34, 42, 43]. Thus, in line with most studies on the subject, our findings support the importance of detecting EDs and starting treatment at an early stage to achieve a better treatment outcome. However, it should be mentioned that also low-weight and medically unstable AN patients have been shown to benefit from FBT, combined with inpatient treatment when indicated [18, 21].

In the literature, early weight gain has been shown to be a prognostic indicator of remission in FBT. We included weight gain by one month of FBT as an assessment parameter, as it has been found to be a strong early predictor of remission in previous studies [17, 26, 28]. In our study, there were no statistically significant differences in weight gain in the first weeks of treatment between those who achieved normal body weight at EOT and those who did not. One possible explanation for this might be in the differences between strict research settings and clinical practice, where the FBT protocol might be more flexible, making it more difficult to assess 'early weight gain' precisely [21]. Participants with a %EBW≥95 at EOT showed significantly higher overall weight gain than those with a %EBW < 95 at EOT. The average weight gain in the first four weeks of FBT was approximately 3 kg in both those with a %EBW < 95 and those with a %EBW≥95 at EOT, which should be prognostic of a good outcome. Thus, the difference in weight gain in treatment overall, which was statistically significant with a large effect size, was apparently due to some characteristics we did not include in the study design (except for baseline underweight severity).

We found no association between illness duration and treatment outcome. In previous research [30, 31], longer

illness duration has been associated with a poorer treatment outcome. However, in recent real-life studies, results comparable to ours have been reported. Goldstein et al. [23] were not able to detect any significant predictors for weight-related treatment outcomes of FBT for adolescent AN in a private practice setting. Likewise, in a recent study on FBT effectiveness in Asian adolescents with AN [24], duration of illness, as well as the patients' age, did not significantly predict the likelihood of weight restoration at 12 months. Further research is needed to better understand the predictors and moderators of clinical FBT outcomes.

In our study, participants with %EBW≥95 at EOT showed significantly higher rates of regular menstrual periods, as well as significantly lower rates of dietary restrictions and cognitive or behavioral symptoms of the ED, at EOT. The result is not surprising. Weight gain emerged as a significant predictor of improved ED pathology and mood symptoms [44, 45]. Based on the findings of numerous studies, the British national guidelines on the recognition and treatment of EDs state that, in AN, weight gain is essential in supporting other psychological, physical, and quality-of-life changes that are required for improvement or recovery [7].

Our results strongly support the current thinking that in the treatment of AN, weight restoration should be substantial enough to achieve significant clinical improvement. Neuroscience research has demonstrated that cerebral gray and white matter deficits due to malnutrition, associated with anxiety, body image and cognitive functions, are reversible with full weight recovery in adolescents with AN [46, 47].

A clinically relevant finding of the current study was that FBT was a sufficient intervention for 42.3% of AN patients admitted to the treatment., while 57.7% required further treatment. Need for additional treatment after FBT has repeatedly been reported in previous research [10, 18]. The 22 participants who did not need further treatment after FBT showed significantly higher weight gain during FBT than those needing further treatment (p < 0.001), though there was no statistically significant difference in %EBW at EOT between these groups. Further, the participants who did not need further treatment after FBT had significantly lower rates of psychiatric comorbidity, less previous psychiatric treatment, and less need for psychopharmacological treatment than participants needing further treatment. At EOT, those who did not need further treatment not only showed lower rates of ED symptoms, but also lower rates of anxiety and depression, in comparison with participants who needed further treatment. In previous research, comorbid psychiatric disorders and prior mental health treatment have likewise been associated with lower remission

rates [30, 31]. We found that nearly half (40%) of patients who needed further treatment after FBT had a comorbid, non-ED psychiatric diagnosis, which is in line with previous studies [48]. These participants exhibited high rates of anxiety and depression symptoms. Indeed, comorbidity demands specific clinical attention in adolescents with AN.

To our knowledge, this is the first study on FBT for AN in Finland. Several challenges in implementing FBT in Finland should be mentioned. In Finland, both parents of an adolescent are often employed full-time, which may limit their ability to provide support at mealtimes and engage in weekly FBT appointments. Young people in Finland are used to a high degree of independence in their daily activities, including meals, from a relatively young age. The idea of transferring the responsibility for daily meals from the adolescent back to parents may cause hesitance in both carers and adolescent patients. Evidence of FBT effectiveness can encourage clinicians to engage families in this type of treatment. On the other hand, there are still few trained FBT therapists in Finland, and the treatment is less familiar to clinicians than other types of ED treatments, such as individual interventions and/or intensive/day/inpatient treatment. Clinicians may feel unsure whether FBT should be the intervention of choice within their day-to-day practice, especially given the complexity of cases and limited resources. Evidence from naturalistic studies can strengthen clinicians' confidence to implement FBT as a first-line treatment for adolescent AN. On the other hand, real-life study reports from different countries and cultures seem to reveal similarities in FBT implementation challenges, recovery rates and predictors of success [23–25].

# Strengths and limitations

The naturalistic context of the present study included both strengths and limitations. A key strength was that the sample was representative of the treatment unit. The family therapist received in-person training in FBT and supervision. A psychiatrist carefully assessed whether FBT was suitable for the participants. In planning the treatment, many factors were taken into account, including malnourishment, ED symptom severity, medical stability, psychiatric comorbidities, as well as family dynamics and parenting style.

Although FBT should be the first-line treatment for adolescent patients with AN, only 52 of 428 (12%) adolescent patients referred to the unit during the study period were admitted to FBT, which is a clear limitation of the study. A few reasons for that could be considered. As EDs are relatively common in Finland [49], Finnish healthcare presumes that before referral to a

tertiary Eating Disorder unit, patients should receive assessment and assistance at the primary care level. That means that most patients referred to the present unit have severe weight deficit and serious ED symptomatology, severe comorbid non-ED psychiatric disorders, and often unsuccessful treatment attempts at a lower care level. Therefore, most patients at the unit are offered an outpatient treatment tailored for their needs, including individual or group interventions focused on weight restoration, eating behaviors, anxiety management, body dissatisfaction or other urgent issues. Family sessions, as well as nutritionist, physiotherapist or pediatrician consultations may be offered in need, and sometimes day- or full in-patient treatment is required. The often severe symptomatology of the patients treated at the unit is an important fact contributing to the relatively small percentage of patients admitted to FBT. It has been noticed in research reviews that patient clinical complexity may preclude clinicians from using FBT [19, 21].

Also, some of the patients were referred to the unit for a consultation visit and continued their treatment elsewhere. Others came when they turned 13 years old to continue treatment started at child psychiatric services where they already had received family therapy in some form, so a different type of treatment was offered to them. Some of the patients were diagnosed with bulimia nervosa and were offered other treatment approaches. Some families declined the offered FBT.

It should also be mentioned that at the time of data collection, the unit only had one family therapist, who also contributed to the unit's functioning through over types of family work. Therefore, limited resources possibly contributed to the strict selection of patients referred to FBT. The current results will guide clinical decision making at the unit, so that more patients, including patients with more severe symptoms, will be offered FBT [18].

Although the number of cases included in this study was modest, one obvious strength is the absence of dropping out from treatment. It is possible that this is related to the thorough assessment of every participant's suitability for FBT, as well as the sufficient psychoeducation given to the families and the high professional skills of the family therapist implementing the treatment.

The data on cognitive or behavioral and mood symptoms of AN at EOT relies on a clinician's free assessment, not on a structured interview or validated measurements. However, we consider our symptom assessments to be reliable because they were conducted by experienced clinicians who were able to detect changes in a patient's state along the treatment process.

#### **Conclusions**

The present study provides encouraging support for the use of FBT as an effective and well-accepted first-line treatment for adolescent AN. The majority of participants achieved normal body weight at end of treatment (EOT), and this was associated with clinical improvement. Higher pre-treatment weight was prognostic of weight restoration at EOT. FBT may be a sufficient intervention for AN patients with less severe underweight and less severe psychiatric comorbidities. Our findings support the importance of early intervention and an active focus on weight restoration in AN patients. The current results show that FBT can be successfully implemented in Finland and suggest that training more ED clinicians in FBT would be beneficial.

#### Abbreviations

AN Anorexia nervosa ED Eating disorder BMI Body mass index

%EBW Percentage of the expected body weight

DSM-5 The Diagnostic and Statistical Manual of Mental Disorders, Fifth

Edition

EOT End of treatment FBT Family-based treatment

ICD-10 International Classification of Diseases

RCT Randomized clinical trial WHO World Health Organization

#### Acknowledgements

We would like to acknowledge the personnel of the Eating Disorder unit.

# **Author contributions**

JS, MN, and SO contributed to the conception and design of the study; JS and SO contributed to the acquisition, analysis, or interpretation of the data; SO drafted the manuscript; JS and MÅ critically revised the manuscript. All authors have read and approved the final manuscript.

#### Funding

Open Access funding provided by University of Helsinki including Helsinki University Central Hospital. This study had no external funding.

### Availability of data and materials

The data are available from the corresponding author (SO) on reasonable request.

## **Declarations**

#### Ethics approval and consent to participate

The study plan was approved by the Ethics Committee of Helsinki University Hospital before the beginning of the data collection. Patients and their guardians received information about the study both orally and in a cover letter. The participants were assured of the confidentiality and anonymity of the data and of the voluntary nature of participation. The patients and their guardians gave written confirmation of their consent for participating in the study and publication of the results.

# **Competing interests**

The authors declare no conflict of interest.

#### **Author details**

<sup>1</sup>Department of Psychiatry, Eating Disorder Unit, University of Helsinki and Helsinki University Hospital (HUS), P.O. Box 282, 00029 Helsinki, Finland. <sup>2</sup>Psychiatric Hospital, City of Helsinki, Nordenskiöldinkatu 20, 00250 Helsinki, Finland. <sup>3</sup>Department of Psychology and Logopedics, Faculty of Medicine, University of Helsinki, Haartmaninkatu 8, P.O. Box 63 00014, Helsinki, Finland. <sup>4</sup>Department of Psychology, Åbo Akademi University, Arken, Tehtaankatu 2, 20500 Turku, Finland.

Received: 5 March 2023 Accepted: 28 August 2023 Published online: 11 September 2023

#### References

- Jenkins PE, Hoste RR, Meyer C, Blissett JM. Eating disorders and quality of life: a review of the literature. Clin Psychol Rev. 2011;31(1):113–21. https://doi.org/10.1016/j.cpr.2010.08.003. (Epub 2010 PMID: 20817335).
- Jones M, Völker U, Lock J, Taylor CB, Jacobi C. Family-based early intervention for anorexia nervosa. Eur Eat Disord Rev. 2012;20(3):e137–43. https://doi.org/10.1002/erv.2167. (Epub 2012 PMID: 22438094).
- Vall E, Wade TD. Predictors of treatment outcome in individuals with eating disorders: a systematic review and meta-analysis. Int J Eat Disord. 2016;49(4):432–3. https://doi.org/10.1002/eat.22518.Erratumfor:IntJEatDisord.2015Nov;48(7):946-71. (PMID: 27084797).
- Lock J. An update on evidence-based psychosocial treatments for eating disorders in children and adolescents. J Clin Child Adolesc Psychol. 2015;44(5):707–21. https://doi.org/10.1080/15374416.2014.971458. (Epub 2015 PMID: 25580937).
- Watson HJ, Bulik CM. Update on the treatment of anorexia nervosa: review of clinical trials, practice guidelines and emerging interventions. Psychol Med. 2013;43(12):2477–500. https://doi.org/10.1017/S003329171 2002620. (Epub 2012 PMID: 23217606).
- American Psychiatric Association. 2006 Treatment of patients with eating disorders, third edition. American Psychiatric Association. Am J Psychiatry. 163(7 Suppl):4–54. PMID: 16925191.
- National Guideline Alliance (UK). (2017) Eating Disorders: Recognition and Treatment. London: National Institute for Health and Care Excellence (NICE); PMID: 28654225.
- Lock J. Family therapy for eating disorders in youth: current confusions, advances, and new directions. Curr Opin Psychiatry. 2018;31(6):431–5. https://doi.org/10.1097/YCO.000000000000451. (PMID: 30063479).
- Lock J, Le Grange D. Treatment Manual for Anorexia Nervosa: A Family-Based Approach. 2nd ed. New York: Guilford; 2013. p. 289p.
- Agras WS, Lock J, Brandt H, Bryson SW, Dodge E, Halmi KA, Jo B, Johnson C, Kaye W, Wilfley D, Woodside B. Comparison of 2 family therapies for adolescent anorexia nervosa: a randomized parallel trial. JAMA Psychiat. 2014;71(11):1279–86. https://doi.org/10.1001/jamapsychiatry.2014.1025. PMID:25250660;PMCID:PMC6169309.
- Le Grange D, Hughes EK, Court A, Yeo M, Crosby RD, Sawyer SM.
  Randomized clinical trial of parent-focused treatment and family-based treatment for adolescent anorexia nervosa. J Am Acad Child Adolesc Psychiatry. 2016;55(8):683–92. https://doi.org/10.1016/j.jaac.2016.05.007. (Epub 2016 PMID: 27453082).
- Le Grange D, Crosby RD, Rathouz PJ, Leventhal BL. A randomized controlled comparison of family-based treatment and supportive psychotherapy for adolescent bulimia nervosa. Arch Gen Psychiatry. 2007;64(9):1049–56. https://doi.org/10.1001/archpsyc.64.9.1049. (PMID: 17768270).
- Le Grange D, Lock J, Agras WS, Bryson SW, Jo B. Randomized clinical trial of family-based treatment and cognitive-behavioral therapy for adolescent bulimia Nervosa. J Am Acad Child Adolesc Psychiatry. 2015;54(11):886–94. https://doi.org/10.1016/j.jaac.2015.08.008.
- Lock J, Agras WS, Bryson S, Kraemer HC. A comparison of short- and long-term family therapy for adolescent anorexia nervosa. J Am Acad Child Adolesc Psychiatry. 2005;44(7):632–9. https://doi.org/10.1097/01. chi.0000161647.82775.0a. (PMID: 15968231).
- Lock J, Le Grange D, Agras WS, Moye A, Bryson SW, Jo B. Randomized clinical trial comparing family-based treatment with adolescent-focused individual therapy for adolescents with anorexia nervosa. Arch Gen Psychiatry. 2010;67(10):1025–32. https://doi.org/10.1001/archgenpsychiat ry.2010.128.PMID:20921118;PMCID:PMC3038846.
- Lock J, Le Grange D, Agras WS, Fitzpatrick KK, Jo B, Accurso E, Forsberg S, Anderson K, Arnow K, Stainer M. Can adaptive treatment improve outcomes in family-based therapy for adolescents with anorexia nervosa?

- Feasibility and treatment effects of a multi-site treatment study. Behav Res Ther. 2015;73:90–5. https://doi.org/10.1016/j.brat.2015.07.015.
- Madden S, Miskovic-Wheatley J, Wallis A, Kohn M, Hay P, Touyz S. Early weight gain in family-based treatment predicts greater weight gain and remission at the end of treatment and remission at 12-month followup in adolescent anorexia nervosa. Int J Eat Disord. 2015;48(7):919–22. https://doi.org/10.1002/eat.22414. (Epub 2015 Jun 17 PMID: 26488111).
- Madden S, Miskovic-Wheatley J, Wallis A, Kohn M, Lock J, Le Grange D, Jo B, Clarke S, Rhodes P, Hay P, Touyz S. A randomized controlled trial of inpatient treatment for anorexia nervosa in medically unstable adolescents. Psychol Med. 2015;45(2):415–27. https://doi.org/10.1017/S003329171 4001573. (Epub 2014 PMID: 25017941; PMCID: PMC4301212).
- Couturier J, Kimber M, Szatmari P. Efficacy of family-based treatment for adolescents with eating disorders: a systematic review and meta-analysis. Int J Eat Disord. 2013;46(1):3–11. https://doi.org/10.1002/eat.22042. (Epub 2012 Jul 23 PMID: 22821753).
- Lock J, Le Grange D. Family-based treatment: Where are we and where should we be going to improve recovery in child and adolescent eating disorders. Int J Eat Disord. 2019;52(4):481–7. https://doi.org/10.1002/eat. 22980. (Epub 2018 Dec 6 PMID: 30520532).
- Accurso EC, Fitzsimmons-Craft EE, Ciao AC, Le Grange D. From efficacy to effectiveness: comparing outcomes for youth with anorexia nervosa treated in research trials versus clinical care. Behav Res Ther. 2015;65:36– 41. https://doi.org/10.1016/j.brat.2014.12.009. (Epub 2014 Dec 23. PMID: 25557396; PMCID: PMC4306615).
- Le Grange D, Gorrell S, Hughes EK, Accurso EC, Yeo M, Pradel M, Sawyer SM. Delivery of family-based treatment for adolescent anorexia nervosa in a public health care setting: research versus non-research specialty care. Front Psychiatry. 2020;22(10):1001. https://doi.org/10.3389/fpsyt. 2019.01001
- Goldstein M, Murray SB, Griffiths S, Rayner K, Podkowka J, Bateman JE, Wallis A, Thornton CE. The effectiveness of family-based treatment for full and partial adolescent anorexia nervosa in an independent private practice setting: Clinical outcomes. Int J Eat Disord. 2016;49(11):1023–6. https://doi.org/10.1002/eat.22568. (Epub 2016 PMID: 27270494).
- Chew CSE, Kelly S, Tay EE, Baeg A, Khaider KB, Oh JY, Rajasegaran K, Saffari SE, Davis C. Implementation of family-based treatment for Asian adolescents with anorexia nervosa: A consecutive cohort examination of outcomes. Int J Eat Disord. 2021;54(1):107–16. https://doi.org/10.1002/ eat.23429. (Epub 2020 Dec 8 PMID: 33290613).
- Simic M, Stewart CS, Konstantellou A, Hodsoll J, Eisler I, Baudinet J. From efficacy to effectiveness: child and adolescent eating disorder treatments in the real world (part 1)-treatment course and outcomes. J Eat Disord. 2022;10(1):27. https://doi.org/10.1186/s40337-022-00553-6.PMID:35189 967;PMCID:PMC8862310.
- 26. Le Grange D, Accurso EC, Lock J, Agras S, Bryson SW. Early weight gain predicts outcome in two treatments for adolescent anorexia nervosa. Int J Eat Disord. 2014;47(2):124–9. https://doi.org/10.1002/eat.22221.
- Couturier J, Lock J. What is recovery in adolescent anorexia nervosa? Int J Eat Disord. 2006;39(7):550–5. https://doi.org/10.1002/eat.20309. (PMID: 16791851).
- Doyle PM, Le Grange D, Loeb K, Doyle AC, Crosby RD. Early response to family-based treatment for adolescent anorexia nervosa. Int J Eat Disord. 2010;43(7):659–62. https://doi.org/10.1002/eat.20764.PMID:19816862; PMCID:PMC8693442.
- Hughes EK, Sawyer SM, Accurso EC, Singh S, Le Grange D. Predictors of early response in conjoint and separated models of family-based treatment for adolescent anorexia nervosa. Eur Eat Disord Rev. 2019;27(3):283–94. https://doi.org/10.1002/erv.2668. (Epub 2019 Feb 13 PMID: 30761665)
- Le Grange D, Lock J, Agras WS, Moye A, Bryson SW, Jo B, Kraemer HC. Moderators and mediators of remission in family-based treatment and adolescent focused therapy for anorexia nervosa. Behav Res Ther. 2012;50(2):85–92. https://doi.org/10.1016/j.brat.2011.11.003. (Epub 2011 Nov 29. PMID: 22172564; PMCID: PMC3260378).
- Lock J, Couturier J, Bryson S, Agras S. Predictors of dropout and remission in family therapy for adolescent anorexia nervosa in a randomized clinical trial. Int J Eat Disord. 2006;39(8):639–47. https://doi.org/10.1002/eat. 20328. (PMID: 16927385).

- Nyman-Carlsson E, Birgegård A, Engström I, Gustafsson SA, Nevonen L. Predictors of outcome among young adult patients with anorexia nervosa in a randomised controlled trial. Eur Eat Disord Rev. 2019;27(1):76–85. https://doi.org/10.1002/erv.2630. (Epub 2018 Aug 9 PMID: 30094893).
- Halvorsen I, Reas DL, Nilsen JV, Rø Ø. Naturalistic outcome of family-based inpatient treatment for adolescents with anorexia nervosa. Eur Eat Disord Rev. 2018;26(2):141–5. https://doi.org/10.1002/erv.2572. (Epub 2017 Dec 8 PMID: 29218761).
- Rosling A, Salonen Ros H, Swenne I. One-year outcome and incidence of anorexia nervosa and restrictive eating disorders among adolescent girls treated as out-patients in a family-based setting. Ups J Med Sci. 2016;121(1):50–9. https://doi.org/10.3109/03009734.2016.1141810. (Epub 2016 Feb 26. PMID: 26915921; PMCID: PMC4812058).
- 35. World Health Organization(WHO). The ICD-10 classification of mental and behavioral disorders: Diagnostic criteria for research. Genève, Switzerland: World Health Organization; 1993.
- Suokas JT, Suvisaari JM, Gissler M, Löfman R, Linna MS, Raevuori A, Haukka J. Mortality in eating disorders: a follow-up study of adult eating disorder patients treated in tertiary care, 1995–2010. Psychiatry Res. 2013;210(3):1101–6. https://doi.org/10.1016/j.psychres.2013.07.042. (Epub 2013 Aug 17 PMID: 23958333).
- Suokas JT, Suvisaari JM, Grainger M, Raevuori A, Gissler M, Haukka J. Suicide attempts and mortality in eating disorders: a follow-up study of eating disorder patients. Gen Hosp Psychiatry. 2014;36(3):355–7. https://doi.org/10.1016/j.genhosppsych.2014.01.002. (Epub 2014 Jan 13. PMID: 24559792)
- Fairburn CG, Beglin SJ. Assessment of eating disorders: interview or selfreport guestionnaire? Int J Eat Disord. 1994;16(4):363–70.
- Beck AT, Ward CH, Mendelson M, Mock J, Erbaugh J. An inventory for measuring depression. Arch Gen Psychiatry. 1961;4:561–71.
- Bohn K, Fairburn CG. The clinical impairment assessment questionnaire (CIA 30). In: Fairburn CG, editor. Cognitive Behavior Therapy and Eating Disorders. New York: Guilford Press; 2008.
- Cohen J. A power primer. Psychol Bull. 1992;112(1):155–9. https://doi.org/ 10.1037//0033-2909.112.1.155. (PMID: 19565683).
- 42. Forman SF, McKenzie N, Hehn R, Monge MC, Kapphahn CJ, Mammel KA, Callahan ST, Sigel EJ, Bravender T, Romano M, Rome ES, Robinson KA, Fisher M, Malizio JB, Rosen DS, Hergenroeder AC, Buckelew SM, Jay MS, Lindenbaum J, Rickert VI, Garber A, Golden NH, Woods ER. Predictors of outcome at 1 year in adolescents with DSM-5 restrictive eating disorders: report of the national eating disorders quality improvement collaborative. J Adolesc Health. 2014;55(6):750–6. https://doi.org/10.1016/j.jadohealth.2014.06.014. (Epub 2014 Sep 4 PMID: 25200345).
- Wild B, Friederich HC, Zipfel S, Resmark G, Giel K, Teufel M, Schellberg D, Löwe B, de Zwaan M, Zeeck A, Herpertz S, Burgmer M, von Wietersheim J, Tagay S, Dinkel A, Herzog W. Predictors of outcomes in outpatients with anorexia nervosa - Results from the ANTOP study. Psychiatry Res. 2016;30(244):45–50. https://doi.org/10.1016/j.psychres.2016.07.002. (Epub 2016 Jul 8 PMID: 27467700).
- Mattar L, Huas C, Duclos J, Apfel A, Godart N. Relationship between malnutrition and depression or anxiety in Anorexia Nervosa: a critical review of the literature. J Affect Disord. 2011;132(3):311–8. https://doi.org/10. 1016/j.jad.2010.09.014. (Epub 2010 Oct 2 PMID: 20920829).
- Accurso EC, Ciao AC, Fitzsimmons-Craft EE, Lock JD, Le Grange D. Is weight gain really a catalyst for broader recovery?: The impact of weight gain on psychological symptoms in the treatment of adolescent anorexia nervosa. Behav Res Ther. 2014;56:1–6. https://doi.org/10.1016/j.brat.2014. 02.006. (Epub 2014 Feb 28. PMID: 24632109; PMCID: PMC4019781).
- Bernardoni F, King JA, Geisler D, Stein E, Jaite C, Nätsch D, Tam FI, Boehm I, Seidel M, Roessner V, Ehrlich S. Weight restoration therapy rapidly reverses cortical thinning in anorexia nervosa: a longitudinal study. Neuroimage. 2016;15(130):214–22. https://doi.org/10.1016/j.neuroimage. 2016.02.003. (Epub 2016 Feb 11 PMID: 26876474).
- Mysliwiec R. Neuroscience of adolescent anorexia nervosa: implications for family-based treatment (FBT). Front Psychiatry. 2020;21(11):418. https://doi.org/10.3389/fpsyt.2020.00418.PMID:32670097;PMCID:PMC73 26098.
- 48. Herpertz-Dahlmann B. Adolescent eating disorders: update on definitions, symptomatology, epidemiology, and comorbidity. Child Adolesc

- Psychiatr Clin N Am. 2015;24(1):177–96. https://doi.org/10.1016/j.chc. 2014.08.003. (**Epub 2014 Oct 7 PMID: 25455581**).
- Silén Y, Sipila PN, Raevuori A, Mustelin L, Marttunen M, Kaprio J, Keski-Rahkonen A. DSM-5 eating disorders among adolescents and young adults in Finland: a public health concern. Int J Eat Disord. 2020;53(5):520–31. https://doi.org/10.1002/eat.23236. (Epub 2020 Jan 30 PMID: 31999001).

## **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

# Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- $\bullet\,\,$  maximum visibility for your research: over 100M website views per year

#### At BMC, research is always in progress.

**Learn more** biomedcentral.com/submissions

